Author’s response to reviews

Title: Assessing Data Availability and Quality Within an Electronic Health Record System Through External Validation Against an External Clinical Data Source

Authors:

Ellen Palmer (ellelpalmer2018@gmail.com; elp76@case.edu)
John Higgins (John.H.Higgins@Dartmouth.edu)
Saeed Hassanpour (Saeed.Hassanpour@Dartmouth.edu)
James Sargent (James.D.Sargent@Dartmouth.edu)
Christina Robinson (Christina.M.Robinson@Dartmouth.edu)
Jennifer Doherty (Jen.Doherty@hci.utah.edu)
Tracy Onega (Tracy.L.Onega@Dartmouth.edu)

Version: 4 Date: 25 Jun 2019

Author’s response to reviews:

To whom it may concern,

We apologize for the error in author contributions - JH has been added on line 387.

We have also removed the text from the consent for publication section and replaced it with "not applicable" as noted.

Thank you once,

Ellen "Elle" Palmer, PhD